Author’s response to reviews

Title: Y-27632 is associated with corticosteroid-potentiated control of pulmonary remodeling and inflammation in guinea pigs with chronic allergic inflammation

Authors:

Patricia Pigati PP (pangelipigati@gmail.com)
Renato F Righetti RFR (renatofragar@yahoo.com.br)
Samantha S Possa SSP (samanthasz@hotmail.com)
Beatriz S Romanholo BSR (baetrizmsaraiva@gmail.com)
Adriana PD Rodrigues APDR (adrianaapalmeira@hotmail.com)
anelize SA dos Santos ASAS (anelize.sartori@hotmail.com)
Débora G Xisto DGX (deboraxisto@gmail.com)
Mariana A Antunes MAA (maryantunes@gmail.com)
Carla M Prado CMP (cmaximoprado@gmail.com)
Edna A Leick EAL (leick51@yahoo.com.br)
Milton de A Martins MAM (mmartins@usp.br)
Patricia RM Rocco PRMR (prmrocco@gmail.com)
Iolanda de FLC Tibério IFLCT (iocalvo@uol.com.br)

Version: 3
Date: 9 May 2015

Author’s response to reviews:

MS: 2016844251139239 – Response to the referees
For the convenience of the referees, we re-typed the questions.

Reviewer’s report

Title: Y-27632 is associated with corticosteroid-potentiated control of pulmonary remodeling and inflammation in guinea pigs with chronic allergic inflammation

Reviewer#1:

The authors should be complimented on a responsive and effective revision of the manuscript which describes a very interesting study.

Minor

1. Table 2, legend. Collagen, elastin and actin are expressed as percentage of what? Please clarify in the legend.
Response: The correct is percentage of the increase. We made the alteration in manuscript.

2. Table 3, legend. PGF2alpha is expressed as percentage of what? Please clarify in the legend.
Response: The correct is percentage of the increase. We corrected it and we thank the referee to point out this mistake.
Reviewer#2:
Major Compulsory Revisions
- The authors have addressed all previously raised points sufficiently well

Minor Essential Revisions
1) Typographical errors
- “INF-#” should be IFN-# (page 28)
Response: We agreed and have made this correction in the manuscript. Now you will find it on page 26.

- Consider changing the legend title for Figure 4 from “Figure 4A, 4B, 4C and 4D” to “Figure 4” to maintain consistency (page 36)
Response: We made the alteration suggested in manuscript. Now you will find it on page 33. Thank you.

Discretionary Revisions
Quality of written English: Needs some language corrections before being published
Response: The manuscript was sent for the American Journal Experts (AJE) for the revision of English.